Author’s response to reviews

Title: The VicGeneration study - a birth cohort to examine the environmental, behavioural and biological predictors of early childhood caries: background, aims and methods

Authors:

Andrea M de Silva-Sanigorski (andreams@unimelb.edu.au)
Hanny Calache (Hanny.Calache@dhsy.org.au)
Mark Gussy (M.Gussy@latrobe.edu.au)
Stuart Daspher (stuartgd@unimelb.edu.au)
Jane Gibson (jgib@unimelb.edu.au)
Elizabeth Waters (ewaters@unimelb.edu.au)

Version: 4 Date: 22 February 2010

Author’s response to reviews:

A few minor typographical errors were amended in the manuscript, specifically:
Pg 11, first paragraph, line 5 should read “used to capture information on oral health related...” ie the ‘to’ is currently missing
Pg 13, There is an extra full stop after reference 28 is cited (at the end of the paragraph)
Pg 14, In the last line of the third paragraph where it states “(tooth 16, 21, 24, 44, 41 and 36)”, there needs to be a space before ‘41’
Pg 14, There is an accent on the ‘u’ in ‘Muhlemann’ on this page – but not on page 25 where the reference is cited.
Pg 15, In the last paragraph, line 4 “lesions ,(decayed), missing and filled...” the space between ‘lesions’ and the comma seems out of place?
Pg 22, Reference 6, the fourth line needs to be tabbed in.
Pg 21, There is an extra dash after “The York review --

All of these have no been fixed in the revised manuscript.

regards
Andrea